Author's response to reviews

Title: Quality of reporting of trial abstracts needs to be improved: using the CONSORT for abstracts to assess the four leading Chinese medical journals of traditional Chinese medicine

Authors:

Ling Wang (linewang84@163.com)
Yulin Li (lee-yulin@163.com)
Jing Li (lijing68@hotmail.com)
Mingming Zhang (mingming-zhang@163.com)
Lin Xu (xulin@163.com)
Wenming Yuan (yuanwenming2000@163.com)
Gang Wang (wong-gang@126.com)
Sally Hopewell (hopewell@cochrane.co.uk)

Version: 5 Date: 8 July 2010

Author's response to reviews:

Minor correction of language error in page 6 and formatting in the table 1.